Author’s response to reviews

Title: Economic support to improve tuberculosis treatment outcomes in South Africa: a qualitative process evaluation of a cluster-randomised controlled trial

Authors:

Elizabeth Lutge (elizabethlutge@gmail.com)
Simon Lewin (Simon.Lewin@kunnskapssenteret.no)
Jimmy Volmink (jvolmink@sun.ac.za)

Version: 3  Date: 23 May 2014

Author’s response to reviews:

Dear Editor

Thank you for your email of 22 May 2014. You requested a few minor changes my manuscript, which I have listed below together with my responses.

Please let me know if these changes are acceptable.

Thanks

1. Please include a figure title and legend section after the reference list. Referee 2 also had some additional recommendations: “Minor essential revisions:

I have put a caption for the figures, as well as for the tables, in bold.

Page 5: Correct the grammatical error in the sentence "In addition, nurse were required to sign each voucher..." Have made the change on page 5

Page 7: Use numerical references for Day et al 2012 and Day et al 2009” Have added numerical references, which has necessitated a change in reference numbering (which I have also made).